EUROPEAN CENTRAL BANK

Working Paper Series

Falk Mazelis, Roberto Motto, Annukka Ristiniemi Monetary policy strategies for the euro area: optimal rules in the presence of the ELB



Disclaimer: This paper should not be reported as representing the views of the European Central Bank (ECB). The views expressed are those of the authors and do not necessarily reflect those of the ECB.

Abstract

We study alternative monetary policy strategies in the presence of the lower bound on nominal interest rates and a low equilibrium real rate using an estimated DSGE model for the euro area. We demonstrate that simple feedback rules that implement inflation targeting result in a binding lower bound one-fourth of the time as well as inflation and output exhibiting large downward biases and heightened volatility. Rule-based asset purchases that are activated once the policy rate reaches the lower bound are not able to fully offset the destabilizing effects of the lower bound if we assume plausible limits on the size of purchases. Makeup strategies, especially average inflation targeting with a long averaging window, perform better than inflation targeting. However, differences in performance across strategies become small if the response coefficients of the feedback rules are optimized. In addition, we find that the benefits of makeup strategies tend to vanish if agents exhibit a degree of inattention to central bank policies as estimated in the data.

Keywords: Monetary policy, effective lower bound, forward guidance, asset purchases, optimal policy, makeup strategies

JEL Classification: E31, E32, E37, E52, E58, E61, E71

Non-Technical Summary

Unlike changes in monetary policy instruments, which are set at policy-meeting frequency, the monetary policy strategy of a central bank is changed rarely. In turn, the setting of policy instruments is guided by the principles established in the monetary policy strategy. This calls for the adoption of a strategy that can adapt to different contingencies.

At the time when the euro area monetary policy strategy was originally designed in the 1990s, the main concern for monetary policy was to anchor inflation at low levels in the face of the inflationary history of previous decades. The period that followed the global financial crisis witnessed a profound change in landscape: the downward trend in the equilibrium real interest rate accelerated and a persistent below-target inflation environment created a concrete risk of de-anchoring of long-term inflation expectations. Addressing low inflation is different from addressing high inflation because the ability of a central bank to provide the necessary accommodation to maintain price stability is constrained by the lower bound on nominal interest rates.

These developments have been common across advanced economies and have generated the urgency to quantify the destabilizing effects of the lower bound on nominal interest rates and to design policy strategies that perform well also in the presence of the lower bound. In practice, inflation targeting – understood as a strategy in which past deviations (undershootings or overshootings) of inflation compared to an inflation target are treated as bygones – is taken as benchmark and the properties of alternative strategies are assessed via model simulations as a sort of laboratory experiment.

We focus on the euro area using an estimated dynamic stochastic general equilibrium model that allows for a trend in the equilibrium real interest rate.

Under the assumption of a 2% inflation target and a level of the real rate in line with estimates for the euro area, we find that Taylor-type policy rules typically taken to represent inflation targeting lead the lower bound to bind one fourth of the time. In addition, inflation and the output gap exhibit large downward biases and heightened volatility.

We find that asset purchases are not able to fully offset the downward biases in inflation and the output gap when we impose plausible limits on the size of asset purchases. We reach similar conclusions when we allow for moderately negative policy rates.

This motivates the search for alternative policy strategies. We focus on a class of policy rules that belongs to makeup strategies – whereby past under(over)-shootings of inflation are not treated as bygones but they influence future policy – and a class that belongs to asymmetric inflation targeting. Within makeup strategies, we consider average inflation targeting (AIT), price level targeting (PLT) and nominal GDP targeting (NGDPT). We

find that makeup strategies perform much better than inflation targeting when strategies are implemented via simple policy rules as typically calibrated in the economic literature. However, these calibrations are somewhat arbitrary. When we optimise the value of the response coefficients in the policy rules, the differences in performance with respect to inflation targeting and across makeup strategies are greatly reduced. *Prima facie* this finding suggests that, in empirically plausible models, makeup strategies are not the only way to improve upon standard policy rules.

In addition, we document that the benefits of makeup strategies rely critically on the credibility of the central bank because in the face of the lower bound such strategies require to make promises of lower-for-longer policy rates. Using empirical estimates proxying for credibility – suggesting that policy actions occurring more than 4-5 years out into the future hardly affect agents' current behaviour – we show that lower-for-longer policies may exert only little traction in an empirically plausible model. The reason is that it requires to make promises extending far into the future. This is especially the case if the new policy is adopted in conditions in which the yield curve is rather flat, a situation that prevailed at the time the ECB decided on its new policy strategy in July 2021.

1 Introduction

At the time when the euro area monetary policy strategy was originally designed in the 1990s the main concern for monetary policy was to anchor inflation at low levels in the face of the inflationary history of the previous decades. The decade that followed the global financial crisis witnessed a profound change in landscape: the downward trend in the equilibrium real interest rate accelerated (see, among others, Holston, Laubach, and Williams, 2017 and Brand and Mazelis, 2019) and a persistent below-target inflation environment created a concrete risk of de-anchoring of long-term inflation expectations. Addressing low inflation is different from addressing high inflation because the ability of the central bank to provide the necessary accommodation to maintain price stability is constrained by the lower bound on nominal interest rates.

These developments have a global dimension and have generated interest in the stabilization properties of alternative policy strategies vis-à-vis inflation targeting, where the latter is understood as a strategy in which past undershootings (or overshootings) of inflation are treated as bygones. For instance, Kiley and Roberts (2017) and Bernanke, Kiley, and Roberts (2019), among others, consider the effects of policy strategies such as average inflation targeting (AIT) and price level targeting (PLT), which share the property that the central bank promises to keep rates lower for longer when the economy hits the lower bound. Most of this literature has limited the analysis to simple feedback rules, has not considered non-standard policy measures, and has focused on performance over the longer run paying lower attention to the prevailing macroeconomic conditions at the time of implementation of a new strategy. We contribute to filling these gaps by making four main contributions to the literature.

First, we estimate a DSGE model for the euro area to study alternative monetary policy strategies in an environment with the lower bound on nominal interest rates and low equilibrium real interest rate. Most of the literature has focused on the U.S. — for contemporaneous and independent work using an estimated model of the euro area instead, see Coenen, Montes-Galdón, and Schmidt (2021) and Erceg, Jakab, and Lindé (2021). The model employed in this study is a medium-scale DSGE model building on Smets and Wouters (2007), extended to include QE and time variation in the equilibrium real rate. The latter is modelled in line with Holston, Laubach, and Williams (2017) as a function of long-run growth and an idiosyncratic component. In addition, we construct the output gap as trend output gap, as this better reflects (revealed) preferences of central banks compared to the flex price output gap of Smets and Wouters (2007).¹ We impose the lower bound and run stochastic simulations to study the implications of different

¹See Adolfson, Laseen, Linde, and Svensson (2011) for a discussion of different notions of the output gap relevant for actual policy making.

values of the inflation target and the equilibrium real rate. Under the assumption of a 2% inflation target and a level of the real rate in line with estimates for the euro area, we find that feedback rules typically taken to represent flexible inflation targeting lead the lower bound to bind one fourth of the time, as well as to inflation and the output gap to exhibit large downward biases and heightened volatility.

Second, we include non-standard monetary policy measures. These instruments have played an important role in actual policy making since the global financial crisis and it may be expected that they can help overcome the destabilizing effects of the lower bound. We represent large scale asset purchases as governed by a rule prescribing to employ asset purchases once the policy rate reaches the lower bound. We find that asset purchases are not able to fully offset the downward biases in inflation and the output gap when we impose plausible limits on the size of asset purchases. We reach similar conclusions when in addition we allow the central bank to set interest rates at moderately negative levels as observed in the euro area in the last few years. We conclude that non-standard measures do not eliminate the need for rethinking the inflation targeting framework.

Third, we compare alternative policy strategies, focusing on average inflation targeting (AIT) with different lengths of the backward averaging window, price level targeting (PLT), nominal GDP targeting (NGDPT) and asymmetric inflation targeting. We implement these strategies via simple feedback rules in the spirit of, among others, Kiley and Roberts (2017) and Bernanke, Kiley, and Roberts (2019), in turn building on earlier work by for instance Orphanides and Wieland (1998), Reifschneider and Williams (2000) and Williams (2009). We find that AIT, PLT and NGDPT perform much better than inflation targeting when strategies are implemented via simple rules as typically calibrated. But we contribute to the literature by computing optimal simple rules and showing that the differences in performance with respect to inflation targeting and across makeup strategies are greatly reduced when we use optimized response coefficients.²

Fourth, while early literature has typically assumed that the policy strategy is fully credible and well understood, see for instance Kiley and Roberts (2017), we contribute to the research pointing to the crucial role of expectations formation in determining the performance of makeup strategies, see Bernanke, Kiley, and Roberts (2019), Amano, Gnocchi, Leduc, and Wagner (2020), Budianto, Nakata, and Schmidt (2020), and Busetti, Neri, Notarpietro, and Pisani (2021).³ We show that policy promises of lower-for-longer embedded in makeup strategies increase in power the more the model suffers from excess sensitivity of inflation to interest rate expectations – the forward guidance puzzle. We find

²Optimal policy in the presence of the lower bound is analysed in stylised models in, among others, Eggertsson and Woodford (2003) and Adam and Billi (2006), and optimal simple rules for temporary price level targeting for the US are studied in Hebden and López-Salido (2018).

³Honkapohja and Kaushik (2020) and Bodenstein, Hebden, and Winkler (2019) study the transition to PLT in a learning environment at the lower bound.

that the degree of inattention to the central bank's promises found in the data — which implies that policy actions occurring more than 4-5 years out into the future hardly affect agents' current behaviour (see de Groot et al., 2021b) — implies that lower-for-longer policies can exert little traction. This is especially the case if the new policy is adopted in conditions in which the yield curve is rather flat, admittedly a situation that prevailed at the time the ECB decided on its new policy strategy in July 2021. When the yield curve is flat at the level of the lower bound over a long horizon, to be able to provide meaningful accommodation via lower-for-longer promises it is necessary to flatten out the yield curve at very long horizons. But at such horizons, policy promises have little impact.

The paper is organised as follows. Section 2 presents the model and the estimation results. Section 3 discusses the methodology we use to compare the performance of alternative policy strategies. Section 4 presents the properties of inflation targeting in an environment with the lower bound and low equilibrium real rate. Section 5 extends the analysis to include non-standard policy measures. Section 6 studies the performance of alternative strategies represented via simple feedback rules. Section 7 computes optimized simple rules and compares their performance. Section 8 computes fully optimal policy and analyses the role played by the excess sensitivity of inflation to interest rate expectations in determining the effects of lower-for-longer promises. Section 9 presents conclusions.

2 Model specification and estimation

The backbone of the model is the DSGE model of Smets and Wouters (2007), which we modify along three dimensions.⁴ First, we add permanent technology shocks that lead to a unit root process for the level of technology. This allows us to define the output gap in terms of deviations of output from its long-run stochastic trend, whereby potential output is the trend output level as in Adolfson, Laseen, Linde, and Svensson (2011). Also, this generates a possible source of time variation in the equilibrium real interest rate (see below). Specifically, the production function of the intermediate firms features both permanent (Z_t) and stationary technology shocks (A_t)

$$Y_t = \varepsilon_t^a (K_t^s(i))^\alpha (Z_t L_t(i))^{1-\alpha} - Z_t \Phi.$$
(1)

Potential output grows at the rate of technological progress $g_{z,t}$

$$ln(Z_t/Z_{t-1}) = g_{z,t} = \rho_{q_z}g_{z,t-1} + \eta_t^{g_z}.$$
(2)

To make the model stationary we detrend all real variables by the level of productivity, Z_t .

⁴The complete model is summarized in Appendix A. It also draws on Maih, Mazelis, Motto, and Ristiniemi (2021).

Second, following Corbo and Strid (2020), we introduce a time-varying equilibrium real interest rate (r_t^*) as a time-varying intercept in the policy rule. We assume it is a function of trend output growth and a reduced-form non-growth component, η_t^n , similar to Holston, Laubach, and Williams (2017)

$$r_t^* = \rho^{rn} r_{t-1}^* + \theta^{rn} g_{z,t-1} + \varepsilon_t^n.$$
(3)

For a high value of ρ^{rn} , this process allows to capture highly persistent movements in interest rates as seen in the data while still reverting to the steady state that is consistent with structural parameters.

Given that the model is log-linearised, in the structural equations we simply replace all instances featuring the interest rate with the gap between the interest rate and the nominal neutral rate R_t^* , that is:

$$R_t^{gap} = R_t - R_t^*, \tag{4}$$

where $R_t^* = r_t^* + \pi_t^*$ and π_t^* is the inflation target, which we will assume in the estimation to be constant and thus we write it as a parameter in what follows.

Monetary policy is assumed to follow a Taylor-type rule such that the annualised nominal interest rate, R_t , is set to respond to deviations of inflation from the inflation target and the trend output gap

$$R_t = R_t^* + \rho^r (R_{t-1} - R_{t-1}^*) + (1 - \rho^r) \left[\alpha_\pi (\pi_t - \pi^*) + \alpha_y y_t \right] + \varepsilon_t^m.$$
(5)

Note that the output gap, y_t , is already detrended with the technological progress Z_t at which potential output grows. Hence y_t can be interpreted as a "Beveridge–Nelson cycle" output gap, which allows inference on medium-term growth prospects by measuring the expected degree of economic growth above and below trend levels (Kiley, 2013). The nominal neutral interest rate enters the policy rule as a time-varying intercept. In the estimation we allow the neutral rate to vary, but in the simulations we shift its level permanently by setting ρ^{rn} to 1, and θ^{rn} to 0 and applying a shock of a magnitude that the equilibrium real rate immediately reaches the desired level, effectively rendering it a parameter.

Third, we introduce central bank asset purchases in a simple way whereby they reduce the interest rate faced by the households. As a result, the interest rate gap in the presence of asset purchases becomes also a function of asset purchases:⁵

⁵This approach is in the spirit of Harrison (2017), who introduces portfolio adjustment costs in a canonical New-Keynesian model where QE operates through the portfolio balance channel, and of shadow

$$R_t^{gap} = R_t - R_t^* + \xi^{qe} qe_t \tag{6}$$

The quantitative easing term and the rule governing asset purchases are described in Section 5.

The model is estimated on euro area data over the sample 1995Q1 to 2020Q1. Since the sample includes the period in which the policy rate has been nearing the lower bound, we use the shadow rate constructed in De Rezende and Ristiniemi (2023) as empirical counterpart to the model's short-term rate. Details on the data are provided in Appendix A.

We divide the model's parameters in two sets. One set comprises all parameters in common with Smets and Wouters (2007), as well as those related to trend output growth and the neutral rate process. A subset of those parameters is kept fixed while the others are estimated using Bayesian methods. The parameters and their values are reported in Appendix B and C.

The second set of parameters is related to the impact of asset purchases. Although these parameters could be estimated together with the other parameters of the model, we calibrate them so as to be in line with the elasticities found in a comprehensive Eurosystem's staff assessment of the impact of asset purchases. In our simulations purchasing 30% of the stock of eligible long-term bonds results to approximately 0.7 percentage point peak increase in inflation.⁶ We see this calibration approach as a way to limit the extent to which our findings about the performance of alternative monetary policy strategies may be driven by parameters for which there is no consensus in the literature.

3 Methodology to quantify the impact of alternative policy strategies

We assess the properties of alternative monetary policy strategies by computing the unconditional probability distributions for inflation, the output gap and the short-term nominal interest rate, in the spirit of Bernanke, Kiley, and Roberts (2019). These distributions are obtained by carrying out stochastic simulations around the non-stochastic

rate models where the effect of asset purchases on the term structure of interest rates is translated to a decline in the short rate.

⁶Specifically, Altavilla, Lemke, Linzert, Tapking, and von Landesberger (2021), who review the estimated impact of the ECB's asset purchases on euro area inflation across a range of studies, find that an asset purchase shock of 10% of GDP leads to a cumulative change in inflation of between 0.3 and 0.7 percentage points (interquartile range) over a three year horizon. At a debt/GDP ratio in the euro area of 86%, as measured at the end of the first quarter of 2020 (see Eurostat news release on 22 July 2020), such a shock corresponds to asset purchases on the order of 12% in terms of share of the stock of sovereign bonds.

steady state, taking into account the non-linearity of the lower bound. We assume that the euro area is hit in each period by disturbances that are randomly drawn from the estimated distribution of shocks. As the sample period extends until 2020Q1, we capture the shocks that occurred during the global financial crisis and the sovereign debt crisis but not the COVID-19 pandemic, which may otherwise inject extreme volatility.

We activate all shocks with the exception of the monetary policy shocks, measurement errors and shocks to the non-growth component of r_t^* —we explain the rationale below. This leaves us with eight active shocks. Given this large number of shocks, we follow Kiley and Roberts (2017) and Bernanke, Kiley, and Roberts (2019) and do not account for the precautionary motive associated with the non-linearity generated by the lower bound.⁷ We implement the lower bound constraint on the policy rate and the size-limit on asset purchases (when activated) using the extended path algorithm (Fair and Taylor, 1983).

We run 2000 simulations of 150 periods discarding the first 50 periods each time. Statistics are computed over all retained simulations.⁸ The simulations are repeated for different values of the inflation target and r^* , with and without non-standard policy measures, and for alternative policy strategies. To ensure comparability, the same sequences of shocks are used across exercises, although with 2000 simulations the specific shock series do not play a relevant role.

To facilitate the comparison across alternative policy strategies, changes in r^* are implemented in a way that does not affect the structural parameters of the economy. Specifically, to compute the implications of different steady-state levels of the real interest rate, the literature has used changes in the discount factor or trend output growth (in models with a trend component), or a combination of those. But changing the value of these parameters may lead to different implications for the model dynamics, which may not be supported by the data and may affect the comparison across specifications. We instead apply a one-time shock to the non-growth component of the equilibrium real rate, and no shocks thereafter.⁹ By setting ρ^{rn} to 1, this adjustment leads to a permanent level shift in the natural rate. A cross-check of this approach to adjust the equilibrium real rate with other ways of adjustment via deep parameters suggests that simulations behave

⁷The computational time increases exponentially with the number of shocks in the model, see Adjemian and Juillard (2016). For an analysis in which agents display precautionary motives, see for instance Maih, Mazelis, Motto, and Ristiniemi (2021), who quantify the impact of the precautionary motive from the lower bound in an estimated DSGE model for the euro area.

⁸For random seeds that embody highly contractionary realisations of shocks, the economy may get stuck in a liquidity trap where the simulation method is unable to establish the outcome of economic dynamics. This is akin to the "contractionary black hole" of Eggertsson and Giannoni (2020). In computing the statistics we remove highly contractionary outcomes that lead to a decline of inflation of more than -10 percentage points below the target, that is, deflationary scenarios of below -8% inflation are excluded. Therefore, our results may underestimate the distortions created by the lower bound.

⁹This is consistent with Holston, Laubach, and Williams (2017), who find that the idiosyncratic component of r_t^* explains a large part of its decline.

similarly to an approach that reduces trend growth or increases the household discount factor (see Figure 9 in the Annex for a comparison), but without having to take a stance on the source of adjustment nor without the risk of changing model dynamics that rely on these two parameters.

4 Stabilization properties of an inflation targeting strategy

We start with assessing the properties of a benchmark policy rule that can be considered as an approximation of flexible inflation targeting because past deviations of inflation from target (shortfalls or overshoots) are treated as bygones:

IT:
$$R_t = \rho(R_{t-1}) + (1-\rho) \left(R^* + \alpha_\pi (\bar{\pi}_t^{(4)} - \pi^*) + y_t^{gap} \right)$$
(7)

where R_t is the short-term real interest rate, and R^* is the (annualised) equilibrium interest rate,¹⁰ $\bar{\pi}_t^{(4)}$ is the annual consumer price inflation rate where $\bar{\pi}_t^{(4)} = \sum_{i=1}^4 \pi_{t-i+1}$, and y_t^{gap} is the output gap. In our benchmark calibration we set the annual inflation target, π^* , equal to 2%, $\rho = 0.85$ and $\alpha_{\pi} = 1.5$. This rule is equivalent to the inflation targeting rule in Kiley and Roberts (2017).

When the policy rate is at the effective lower bound (ELB), Equation (7) describes the shadow interest rate, R_t^{Taylor} , that would prevail were the policy rate not restricted by the lower bound. The actual interest rate faced by the economic agents is instead R_t :

$$R_t^{Taylor} = \rho(R_{t-1}^{Taylor}) + (1 - \rho) \left(R^* + \alpha_\pi (\bar{\pi}_t^{(4)} - \pi^*) + y_t^{gap} \right),$$
(8)
$$R = \max(-ELB, R_t^{Taylor}).$$

The lower bound is imposed at 0% and it is assumed that the only monetary policy instrument available to the central bank is the short-term interest rate—these assumptions are relaxed in the following sections.

Figure 1 displays the stabilisation properties of the inflation targeting rule for $\pi^* = 2\%$ and different levels of r^* . Assuming that r^* is 2%, the frequency of hitting the zero lower bound is 11% and the average duration of the lower bound is 17 quarters.¹¹ The lower bound generates a downward bias in inflation and the output gap of 0.1 percentage points and 0.3 percent, respectively. It also leads to heightened volatility.

¹⁰For simplicity we rewrite R_t^* as a parameter in all the equations in this section because in the simulations we add a shock to the non-growth component of r_t^* in the first period, and none thereafter, such that r_t^* and thus R_t^* are shifted permanently to the desired level.

 $^{^{11}\}mathrm{See}$ Table 5 in the Annex for results in tabular form.



Figure 1: Impact of alternative levels of r^* assuming $\pi^* = 2\%$

Note: The figure displays statistics for the lower bound incidence as well as the bias in and the standard deviation of inflation and the output gap. The inflation bias is computed as the difference between mean inflation and the inflation target. The steady-state distributions are computed from stochastic simulations around the non-stochastic steady state. The ELB is set at 0%.

With r^* equal to 1% the frequency of hitting the zero lower bound increases to 18% and the average duration of the lower bound becomes 23 quarters. The downward bias in inflation and the output gap is 0.3 percentage points and 0.8 percent, respectively. Assuming that r^* is equal to 0%, which is consistent with recent estimates for the euro area (Brand, Bielecki, and Penalver, 2018), the frequency of hitting the zero lower bound rises to almost a quarter of the observations, and the average duration of the lower bound increases to 27 quarters. Inflation falls short of the target by 0.5 percentage points and the output gap by 1.3 percent. The standard deviation of inflation and the output gap increases the lower the real interest rate is: when r^* is 0%, it becomes about twice as large compared to the case in which r^* is 2%.

An increase in the inflation target can remedy the distortions introduced by the zero lower bound, while a reduction makes them worse. Figure 2 shows that an increase in the inflation target to 3% would result in similar macroeconomic outcomes as the case where r^* increases from 1% to 2%. Assuming an inflation target at 1% and r^* at 1% delivers a downward bias in inflation close to 0.6 percentage points and in the output gap to about 1.5 percent.¹²

 $^{^{12}}$ The results for an inflation target at 1% should be treated with caution because a significant number of simulations do not converge due to the more extreme outcomes they produce. Therefore, the statistics we report underestimate the distortionary effects of the lower bound.



Figure 2: Impact of alternative levels of the inflation target, assuming $r^* = 1\%$

Note: The figure displays statistics for the lower bound incidence, inflation and the output gap. The inflation bias is computed as the difference between mean inflation and the inflation target. The steady-state distributions are computed from stochastic simulations around the non-stochastic steady state. The ELB is set at 0%.

5 Allowing for non-standard measures

In this section we analyse whether non-standard measures can offset the distortionary effects of the zero lower bound. We focus on negative interest rates and large scale asset purchases.

As regards negative rates, we allow the interest rate to be cut to -0.5%. We select this value as it is the lowest level to which the ECB has cut its interest rate to date. Figure 3 (blue bars) shows the effect of allowing for negative rates assuming an inflation target of 2%. We set the level of r^* to 0.5%, which is in the range of recent estimates of r^* for the euro area. Lower values of r^* would exacerbate further the destabilisation effects of the lower bound under an inflation targeting strategy. We find that the inflation bias is about 0.3 percentage points and the bias in the output gap is 0.7 percent. This result exemplifies that allowing for (mildly) negative rates alone is not sufficient to counteract the adverse effects of the lower bound, and the central bank would need to use additional non-standard measures such as large scale asset purchases to deliver on its target.

As regards large scale asset purchases, we consider three alternative QE rules. In all cases the central bank's balance sheet evolves according to an AR(2) process. This process allows for a gradual build-up of the central bank's portfolio, mimicking the practice typically followed by central banks to make an initial announcement about the size of the purchases while building up the portfolio gradually over time. We define QE as purchased assets as a share of long-term EA bonds. The first rule allows for asset purchases to be conducted also when the interest rate is above the lower bound. We label this rule "permanent QE". The amount of asset purchases is assumed to be a function of the prevailing inflation gap:

$$qe_t = \rho^{qe_1} qe_{t-1} - \rho^{qe_2} qe_{t-2} - \theta^{qe} (\bar{\pi}_t^{(4)} - \pi^*).$$
(9)

The second rule features asset purchases only once the policy rate reaches the lower bound. We label this rule "QE at ELB". The amount of purchases depends indirectly on inflation and the output gap via their impact on the interest rate rule, reminiscent of the state-dependent asset purchasing rule in Coenen, Montes-Galdón, and Smets (2020). However, when the interest rate reaches the ELB, the Taylor rule would call for additional rate cuts if inflation and output fall short of their targets, opening up an interest rate gap. The lower the Taylor rule interest rate is relative to the ELB, the more asset purchases the central bank carries out:

$$qe_t = \rho^{qe_1} qe_{t-1} - \rho^{qe_2} qe_{t-2} + \tau^{qe} max(0, R_t - R_{t-1}^{Taylor}).$$
(10)

The third rule is similar to the second one but we impose a limit to the size of purchases that the central bank can carry out. We set the limit to 50% of the stock of bonds. We label this rule "QE at ELB (50% limit)":

$$qe_t = min(50, \rho^{qe1}qe_{t-1} - \rho^{qe2}qe_{t-2} + \tau^{qe}max(0, R_t - R_{t-1}^{Taylor})).$$
(11)

In the simulations we assume an inflation target of 2%, a level of r^* of 0.5% and a level of the ELB at -0.5%.

Figure 3 shows that if we allow the central bank to employ asset purchases once the interest rate reaches the ELB and we do not impose any limit to the size of the central bank's asset holdings (Equation (10)), we find that average inflation bias is 0.1%, see grey bars.¹³

If we impose an asset-purchase limit at 50% of the stock of bonds, as specified in the third rule (Equation (11)), the inflation bias is 0.1%. This however represents an underestimation of the true inflation bias because many runs do not converge compared to the case where there is no upper bound to asset purchases. Were we to include only the same converged simulations in both cases, inflation bias with a 50% limit on asset purchases would be four basis points lower than without the limit.

 $^{^{13}\}mathrm{See}$ Table 6 in the Annex for results in tabular form.



Figure 3: Impact of non-standard measures, assuming $\pi^* = 2\%$ and $r^* = 0.5\%$

Note: The inflation bias is computed as the difference between mean inflation and the inflation target. The steady-state distributions are computed from stochastic simulations around the non-stochastic steady state. NIRP stands for negative interest rate policy, assumed to allow the policy rate to reach -0.5%. QE is modelled as an endogenous rule that is activated: (i) at the lower bound and with no purchase limits (labelled in the figure as QE at ELB (no limit)), or (ii) at the lower bound but with a purchase limit at 50% (labelled as QE at ELB (50% limit), or is available at all times (labelled as Permanent QE).

To further investigate the role played by imposing limits to the size of QE, Figure 4 shows the distribution of the size of asset purchases as a percentage of the stock of euro area bonds in the case of no limit to purchases (LHS panel) as well as the case with 50% limit (RHS panel). In the large majority of cases, the purchases are below the 50% limit. However, there are instances in which very large purchases would be required even in excess of 100%. These results show that NIRP & asset purchases activated once the policy rate reaches the ELB are not sufficient to overcome the downward inflation bias under plausible parameterizations of the purchasable amounts.



Figure 4: Histograms of QE with and without upper limit of 50%

If instead we allow the central bank to employ asset purchases also away from the ELB and we do not impose size limits to the amount of purchases, as specified in the rule in Equation (9), Figure 3 shows that asset purchases in combination with NIRP can almost fully offset the inflation bias (yellow bars). However, making QE a standard policy tool to be used also in normal times is a step that no central bank has taken so far.

6 Alternative monetary policy strategies: evaluation based on simple policy rules

Given our finding that under an inflation targeting strategy non-standard measures would not be able to address the distortions caused by the ELB if plausible limits are imposed, in this section we assess the stabilization properties of alternative monetary policy strategies. We consider two types of strategies and we abstract from asset purchases.

The first type rests on dropping the assumption that past inflation shortfalls (overshoots) are bygones. Differently from inflation targeting (IT), inflation under-(over-)shooting would then call on the central bank to aim for inflation above (below) the target for some time in the future. This practice has the potential to be especially useful at the lower bound because the central bank cannot provide additional easing via further cuts of the policy rate at that moment whereas it can promise additional easing to be delivered in the future when the lower bound episode ends. We consider average inflation targeting (AIT), price-level targeting (PLT) and nominal GDP targeting (NGDPT). We follow the approach of Bernanke, Kiley, and Roberts (2019) and Chung, Gagnon, Nakata, Paustian, Schlusche, Trevino, Vilán, and Zheng (2019), among others, in specifying the policy strategy in terms of a simple policy rule, whereby an additional term is added to the inflation targeting rule.

Under AIT the average inflation gap is computed over the previous T years:

AIT:
$$R_t = \rho R_{t-1} + (1-\rho) \left(R^* + \alpha_\pi (\bar{\pi}_t^{(4)} - \pi^*) + y_t^{gap} + \alpha_{AIT} (\bar{\pi}_t^{(4T)} - \pi^*) \right)$$
(12)

where the superscript T on $\bar{\pi}_t^{(4)}$ is the length of the averaging window in years. We set $\rho = 0.85$, as in our IT rule, $\alpha_{\pi} = 1$, and $\alpha_{AIT} = T$. In our simulations we consider two alternative windows, T = 4 years and T = 8 years.

The PLT strategy can be seen as a form of AIT where the length of the averaging window is infinitely long:

PLT:
$$R_t = \rho R_{t-1} + (1-\rho) \left(R^* + \alpha_\pi (\bar{\pi}_t^{(4)} - \pi^*) + y_t^{gap} + \alpha_{PLT} (p_t - p_t^*) \right)$$
(13)

where p_t is the price level, and p_t^* is the price-level target, such that $p_t^* = p_{t-1}^* + \pi^*$. We set $\rho = 0.85$, $\alpha_{\pi} = 1$, and $\alpha_{PLT} = 1$.

Under NGDPT the central bank responds to the gap between the actual and the target nominal GDP paths, where the target is based on deterministic real growth, estimated in euro area data to be 1.1% annually, and the price-level target:

NGDPT:
$$R_{t} = \rho R_{t-1} + (1-\rho) \left(R^{*} + \alpha_{\pi} (\bar{\pi}_{t}^{(4)} - \pi^{*}) + y_{t}^{gap} + \alpha_{NGDPT} (y_{t} + p_{t} - (y_{t}^{*} + p_{t}^{*})) \right)$$
(14)

where $y_t + p_t$ is nominal GDP and $y_t^* + p_t^*$ is the target path for nominal GDP. We set $\rho = 0.85$, $\alpha_{\pi} = 1$, and $\alpha_{NGDPT} = 1$.

If the length of the averaging window is equal to 1, AIT collapses to IT, except that a positive value of the coefficient in front of the additional AIT term would increase the overall responsiveness of monetary policy to inflation. This highlights that the comparison across strategies carried out using simple rules may be unsatisfactory. It remains unclear the extent to which better performance of, say, AIT compared to IT is due to a stronger response to inflation under AIT, irrespective of the makeup element, or is indeed due to the makeup element, or a combination of the two. We nevertheless follows this approach because it is the typical one followed in the literature, but we relax it in the next section.

The second type of policy strategy that we consider is asymmetric inflation targeting (aIT). The policy response to inflation is made stronger when inflation is below target. In the absence of the lower bound, this generates an upward inflation bias. However, in the presence of the lower bound it has the potential to offset the downward bias in inflation generated by the lower bound:

aiT:
$$R_{t} = \rho R_{t-1} + (1-\rho) \left(R^{*} + y_{t}^{gap} + \alpha_{\pi} (\bar{\pi}_{t}^{(4)} - \pi^{*}) + \mathbf{I}_{\bar{\pi}_{t}^{(4)} < \pi^{*}} \alpha_{aiT} (\bar{\pi}_{t}^{(4)} - \pi^{*}) \right)$$
(15)

We set $\rho = 0.85$, $\alpha_{\pi} = 1.5$, and $\alpha_{aiT} = 1$.

We evaluate the stabilization properties of the policy strategies using stochastic simulations and employing the same sequences of shocks we have used to evaluate IT. We set $\pi^* = 2\%$, $r^* = 0.5\%$ and we impose the ELB at -0.5%.

Figure 5 demonstrates that all the alternative strategies help reduce the distortions generated by the lower bound and perform better than IT.¹⁴ If the makeup approach is well understood by the public and agents are forward looking as in our model, the expectation that the central bank will provide in the future additional easing that will lead inflation to overshoot to compensate for past inflation shortfalls can have an immediate accommodative effect. PLT and NGDPT deliver the lowest inflation bias across the alternatives. However, PLT generates relatively high volatility in the output gap. The reason is that monetary policy has to tighten significantly to compensate for the higher

 $^{^{14}\}mathrm{See}$ Table 7 in the Annex for results in tabular form.

price level caused for instance by a markup shock. The 8-year AIT rule fares better than PLT in terms of volatilities, while it is only marginally worse in terms of biases. The 4-year AIT rule performs much worse than the 8-year AIT both as regards biases and volatilities. The performance of the asymmetric policy rule is similar to the 4-year AIT. In terms of frequency of hitting the ELB and average duration of ELB episodes, the lowest values are reached by PLT and the highest by NGDPT.

Figure 5: Impact of alternative monetary policy strategies, assuming $\pi^* = 2\%$ and $r^* = 0.5\%$



Note: The figure displays statistics for inflation and the output gap. The inflation bias is computed as the difference between mean inflation and the inflation target. The steady-state distributions are computed from stochastic simulations around the non-stochastic steady state. The ELB is set at -0.5%.

To rank the policy strategies we summarize their performance using the period loss function:

$$L = (\pi_t^4 - \pi^*)^2 + \omega_y (y_t - y^*)^2 + \omega_R (\Delta R_t)^2.$$
(16)

We set the weight on output, ω_y , at 0.25 and the weight on smoothing, ω_R , at 1 as is standard in the literature.¹⁵

Figure 6 shows that NGDPT performs best, followed by 8-year AIT. Even though PLT and aIT improved upon IT in terms of bias reduction, the increased volatility weighs on their desirability in terms of welfare losses. As a caveat, all makeup strategies rely heavily on the expectations channel whereby agents fully anticipate that inflation undershooting will be followed by overshooting. In Section 8 we document the implications of varying the strength of the expectations channel.

7 Optimized simple rules

While in the previous section we have assessed the performance of simple policy rules calibrated as in Bernanke, Kiley, and Roberts (2019), in this section we choose the parameter values of the policy rules optimally following the approach of Levin, Wieland,

¹⁵See Kiley and Roberts (2017) for example.



Figure 6: Welfare losses: comparison across alternative policy strategies

Note: Welfare is based on the results in Table 7. The period loss function is $L = (\pi_t^4 - \pi^*)^2 + \omega_y (y_t - y^*)^2 + \omega_R (\Delta R_t)^2$, with $\omega_y = 0.25$, and $\omega_R = 1$.

and Williams (1999), Levin, Wieland, and Williams (2003) and Williams (2003) among others.¹⁶

The central bank's period loss takes the same form as in the exercise above. We focus on the policy rules describing average inflation targeting (AIT8) and price level targeting (PLT) because in the previous section we have found that they are among the best and worst performing policy strategies, respectively, making them prime candidates to explore the potential of optimized rules.¹⁷ We analyse also IT as it acts as a benchmark strategy.

For each policy strategy, the functional form of the policy rule is the same as the one considered in the previous sections, but we choose optimally the magnitude of some of the parameters. Specifically, we consider two cases. In the first case we compute the optimal value of the response to the inflation (price) gap — namely, the gap between inflation and the inflation objective under IT, the gap between average inflation and the inflation objective under AIT8, and the gap between the price level and the deterministic price trend under PLT. In the second case we choose optimally for each policy rule also the value of the response to the lagged interest rate.

¹⁶In contrast to this early work on optimal simple rules, we impose the lower bound.

¹⁷We do not consider NGDPT here because such a strategy seems to apply more naturally to a dual-mandate central bank than to a central bank with a primary objective as the ECB.

The analysis is based on stochastic simulations using the same sequences of shocks we have used in the previous sections. In the simulations it is assumed that the inflation target is 2%, $r^* = 0.5\%$ and the ELB is -0.5%.

The findings are presented in Table 1. The results for the baseline parameterization employed in the previous sections is shown in the columns labelled 'Baseline', while results for the optimally-computed policy response are displayed in the columns labelled 'Optimal'. Two results stand out. First, in the case in which we choose optimally the response to only one parameter (inflation gap in IT, average inflation gap in AIT8 and price level gap in PLT), under AIT8 and PLT there is hardly any improvement in the loss compared to the baseline parametrization employed in the previous section. On the contrary, under IT there is a very large improvement compared to the baseline parameterization of the IT rule. Evidently, under AIT8 and PLT the optimal response coefficients are close to the values used in the baseline parameterization. Instead, under IT the optimal response to inflation is much stronger than in the baseline parameterization.¹⁸ Both in the baseline parameterization and the optimization of the repose to the inflation gap, AIT8 is the best performing strategy.

	1	Baselin	0		Optimal				
	I IT	AIT8	PLT	IT	AIT8	PLT	IT	AIT8	PLT
				$\alpha_{IT}^{opt} = 5.0$	$\alpha_{AIT}^{opt} = 5.9$	$\alpha_{PLT}^{opt} = 1.4$	$\alpha_{IT}^{opt} = 5.4$	$\alpha_{AIT}^{opt} = 6.2$	$\alpha_{PLT}^{opt} = 1.3$
					All	T LI	$\rho^{opt} = 0.95$	$\rho^{opt} = 0.9$	$\rho^{opt} = 0.8$
LB frequency	18.02	14.27	13.34	21.63	14.05	13.72	12.96	11.36	13.20
LB duration	22.53	20.21	19.30	23.47	20.04	19.71	18.73	18.90	19.11
Mean inflation	1.74	1.97	1.99	1.81	1.97	1.99	1.99	1.98	2.00
Mean ygap	-0.74	-0.21	-0.17	-0.58	-0.21	-0.17	-0.18	-0.15	-0.16
Mean rate	2.87	2.63	2.64	4.36	2.63	2.64	2.63	2.60	2.64
RMSD inflation	3.05	1.39	3.48	1.97	1.43	3.45	1.37	1.38	1.39
RMSD ygap	5.84	4.60	7.49	4.40	4.53	7.55	4.79	4.56	4.39
RMS rate changes	0.41	0.34	0.33	0.58	0.22	0.33	0.33	0.27	0.32
Loss	18.02	7.35	26.29	9.04	7.30	26.22	7.73	7.18	6.84

Table 1: Optimal simple rules under alternative policy strategies and their properties

Note: RMSD is the root mean squared deviations of inflation and the output gap from their targets. RMS is the root mean squared changes in the policy rate. The inflation target is set at 2%, r^* at 0.5% and the ELB at -0.5%.

Second, by optimally choosing also the interest-rate smoothing coefficient, we find that there is an improvement across all policy strategies but especially under PLT. Under IT the improvement is achieved via a stronger inflation response compared to the baseline and stronger interest rate smoothing compared to the baseline. Under AIT8 (PLT) the optimal response to the average inflation (price level) gap is similar to the case in which we optimize only over this term while the optimal interest-rate smoothing coefficient is larger under AIT8 and lower under PLT compared to the baseline parameterization. The

¹⁸The strong inflation response in the optimised simple rule is discussed, for instance, in Levin, Wieland, and Williams (1999), Levin, Wieland, and Williams (2003), and Schmitt-Grohe and Uribe (2007) in a setting that does not consider the ELB.

most dramatic improvement in terms of losses is found for PLT that now emerges as the best performing strategy. Overall, the main finding is that all strategies perform rather well in terms of biases and volatility and the differences in their associated loss become very small.

8 Optimal policy and expectations channel

Having analysed in the previous section optimal simple instrument rules, in this section we turn to fully optimal monetary policy. We take the perspective of standing in mid-2020—a time in which the ECB was undertaking the review of its monetary policy strategy—and inflation was well below target and the interest rate stood at -0.5%. We compute optimal monetary policy and assess the sensitivity of the benefits of makeup strategies to varying the strength of the expectations channel.

We assume that we stand in 2020Q2 and the "baseline projections" for the euro area are the ones provided in the June 2020 Broad Macroeconomic Projection Exercise (BMPE) conducted by Eurosystem staff, extended as follows. The June 2020 BMPE covers the horizon only up to end-2023, hence we extent it using survey information from Consensus Economics for inflation and market-based forward-rates for the short-term interest rate, and we condition on them to generate the path of the other variables. The baseline projections are depicted with the dashed line in Figure 7. We make three observations. First, at the beginning of the projection horizon, in 2020Q2, inflation is 0.2% and by then it had been below target (in annual terms) for seven consecutive years. Second, the baseline projection for inflation foresees a rise towards the 2% target very slowly, still falling short of the target in 2026. Third, the baseline path for the interest rate is very shallow, remaining at -0.5% until mid-2021 and then rising by only 40 basis points by the end of 2026.



Figure 7: Optimal policy with alternative assumptions for private-sector attention

Note: Optimal policy based on the loss function in Equation (17). Results in percentage points. The simulations are conducted around the extended June 2020 BMPE baseline.

Given these baseline projections, we compute the interest rate path that would minimise the loss function¹⁹

$$L_0 = \sum_{t=0}^{\infty} \beta^t \left[(\bar{\pi}_t^{(4)} - \pi^*)^2 + \omega_y (y_t - y^*)^2 + \omega_{dr} (\Delta R_t)^2 \right]$$
(17)

As in the previous section, we set $\omega_y = 0.25$, $\omega_{dr} = 1$, and the ELB at -0.5%. Optimal policy is derived under commitment. We assume that the central bank abstains from using asset purchases.

The yellow line in Figure 7 represents the optimal interest rate path.²⁰ It calls for keeping the interest rate at the ELB for longer, one additional year compared to the baseline. This leads to an inflation overshooting in the future, which in turn lowers the real interest rate and provides an immediate boost to inflation.²¹ As a result, inflation reaches the 2% target already by 2021Q1, several years earlier than in the baseline projections. It is worth noting that this can be achieved with an overshooting of inflation that is very small and short-lived.

The strong and fast inflation response to the expectations of a lower-for-longer monetary policy suggests that the expectations channel is powerful in the model. To quantify the relevance of the expectations channel we vary the extent to which agents internalise the central bank's promise to deliver future accommodation, using the approach of de Groot and Mazelis (2020). Specifically, we assume that a constant fraction of agents are inattentive to central bank announcements about the future course of policy.²² The benchmark case is a situation in which all agents are attentive, and this corresponds to the yellow line in Figure 7.

Figure 7 displays the results for different shares of agents' attentiveness. We make three observations. First, even small deviations from the benchmark case of full attentiveness affect the outcomes significantly. For instance, if the share of attentiveness drops to 90% (grey line in Figure 7), the lift-off date is delayed by more than one year compared to the

¹⁹Svensson (2020) refers to this approach of computing optimal policy as forecast targeting whereby a policy rate path is chosen so that the forecasts of target variables (e.g. inflation and economic activity) "look good", meaning that they minimise the loss function. Differently from optimized simple rules, this approach has the advantage of delivering fully optimal (but thus more complex) policy while retaining transparency and being easy to communicate. We let t = 0 denote the quarter in which the central bank chooses the optimal path for the policy instrument, defined as the path which minimizes the expected value of a specified loss function at quarter 0.

²⁰The simulations are conducted with the impulse response function-based approach described in de Groot, Mazelis, Motto, and Ristiniemi (2021a) to construct conterfactual policy scenarios.

²¹For the history dependence of optimal policy in the presence of the lower bound, see Eggertsson and Woodford (2003), Adam and Billi (2006) and Nakov (2008).

 $^{^{22}}$ We do not model a situation in which there is a switch to a new policy strategy and agents have to learn the new strategy, e.g. via Bayesian updating. We assume that the strategy has been in place for some time and yet a fraction of agents remains inattentive.

case of full attentiveness. In turn, inflation reaches its 2% target two years later than in the case of full attentiveness.

Second, the impact of a declining share of attentive agents is non-monotonic.²³ Specifically, lowering the share of attentiveness leads to optimally postpone the lift-off date, but up to a critical value beyond which lift-off is brought forward again. For instance, if the share of attentive agents is 80% (orange line in Figure 7), lift-off is postponed compared to the case in which the share is 90%. Importantly, although lift-off is postponed, the inflation response is smaller and more delayed compared to the case of 90% attentiveness. If the share of attentiveness drops to 70% (red line in Figure 7) we have a reversal, with lift-off occurring at an earlier date than in the case in which attentiveness is 80%. Notably, with 70% attentiveness, inflation is not much affected by the anticipation of future accommodation. The earlier lift-off when attentiveness drops below a critical value is explained by the fact that in such circumstances policy accommodation in the distant future does little to close the currently prevailing negative inflation gap, which means that it brings little benefit to communicate a longer stay at the ELB already now. However, given the assumed loss function penalizing deviations of inflation from its target, when the future comes and the central bank has to deliver the additional policy accommodation that was promised in the past, it will be very costly. While it is always the case that optimal policy under commitment is time inconsistent, ex ante the cost associated with a future inflation overshooting is worth to be paid only if the additional future accommodation helps to close the negative inflation gap currently prevailing. But this is not the case if the share of attentiveness declines below a critical value.

Third, the effects of varying the share of attentive agents depend on the length of the ELB episode and the steepness of the interest rate path after lift-off as embedded in the baseline projections. The reason is that at the ELB the central bank can provide additional policy accommodation only by promising to postpone lift-off or to hike rates afterwards in a more gradual manner — we are assuming here that the central bank does not employ large-scale asset purchases. However, if the baseline projections embed a lift-off date far out in the future and subsequently the interest rate is foreseen to increase very gradually, additional policy accommodation can only be provided by promising lower interest rates in the very distant future — only at those horizons there is policy space available. But unless there is a high degree of attentiveness, the private sector becomes insensitive to policy actions that will take place in the very distant future. de Groot, Mazelis, Motto, and Ristiniemi (2021b) estimate the share of attentiveness to lie within the range of 70-81%.²⁴ Figure

²³This non-monotonic behaviour was first uncovered in Nakata, Ogaki, Schmidt, and Yoo (2019) in a 3-equation New-Keynesian model using a different mechanism for dampening of FG effects in which agents discount future conditions more strongly.

²⁴de Groot, Mazelis, Motto, and Ristiniemi (2021b) estimate the share of attentiveness together with the other parameters of the DSGE model. They show that using the estimated share of attentiveness their

7 (blue and orange lines) shows that such a share of attentiveness would imply that the benefits in terms of inflation stabilization that can be achieved by keeping the interest rate lower-for-longer as called for by makeup strategies are very modest. At the same time, it would still be optimal to delay lift-off by about one year compared to what is embedded in the baseline projections.

We conclude this section by analyzing Svensson's (2020) proposal of representing averageinflation targeting (price-level targeting) by means of a central bank loss function in which the deviation of the period-by-period inflation rate from its target is replaced by the gap between the moving average of inflation (the price level) and the inflation target (the deterministic price trend):²⁵

AIT:
$$L_0 = \sum_{t=0}^{\infty} \beta^t \left[(\bar{\pi}_t^{(8T)} - \pi^*)^2 + \omega_y (y_t^{gap})^2 + \omega_{dr} (\Delta R_t)^2 \right]$$
(18)

PLT:
$$L_0 = \sum_{t=0}^{\infty} \beta^t \left[(p_t - p_t^*)^2 + \omega_y (y_t^{gap})^2 + \omega_{dr} (\Delta R_t)^2 \right]$$
(19)

The results are presented in Figure 8. First, optimal policy calls for offsetting the large average inflation gap (and the even larger price level gap) accumulated by the time in which we carry out our optimization exercise in 2020Q2. This requires a large inflation overshooting, as shown by the yellow line representing optimal policy assuming agents are fully attentive. Second, a lower degree of attentiveness leads to postpone lift-off,²⁶ and the impact on inflation becomes smaller and more delayed compared to full attention.

These results further exemplify that the benefits of makeup approaches depend on the strength of the expectations channel. For values of attentiveness consistent with empirical estimates, the benefits of makeup strategies tend to be small as shown by the blue and orange lines.

9 Conclusion

We estimate a model on euro area data and study the stabilization properties of alternative policy strategies in the presence of the lower bound and a low equilibrium real rate. If

model does not exhibit the forward guidance puzzle described in Del Negro, Giannoni, and Patterson (2015). Their estimated share of attentiveness implies that agents do not internalise central bank's promises about future policy rates beyond 4-5 years out into the future.

²⁵Selecting the appropriate relative weight on output stabilization across loss functions representing alternative policy strategies is not straightforward (see Vestin, 2006, Svensson, 2020, and Budianto et al., 2020). For simplicity we use the same weight across specifications of the loss function, with $\omega_y = 0.25$.

²⁶This is depicted in all cases except for PLT under full attention. In this scenario, instead of remaining at the ELB for longer, the optimal policymaker employs the power of the FG puzzle by reducing the policy rate by a small amount far in the future (not shown in the chart), which is fully anticipated by forward-looking agents and leads to the large inflation overshoot seen in the near-term.



Figure 8: Optimal policy under Average Inflation Targeting and Price Level Targeting with alternative assumptions for private-sector attention

Note: Top panels: optimal policy based on loss function in Equation (18). Bottom panels: optimal policy based on loss function in Equation (19). Results in percentage points. The simulations are conducted around the extended June 2020 Eurosystem staff baseline.

strategies are represented in terms of simple feedback rules as typically calibrated in the literature, we find that makeup strategies perform much better than inflation targeting, and within makeup strategies we find a clear ranking.

However, we find that differences across strategies become small if we optimize the response coefficients of the rules. Future research should assess the robustness of those rules across a variety of models featuring a low equilibrium real rate and an occasionally-binding lower bound constraint, given that optimized rules may tend to be model specific.

We show that addressing the excess sensitivity of inflation to interest rate expectations (the forward guidance puzzle) has implications for lower-for-longer policies such as makeup strategies. Assuming a degree of agents' attention to central bank's promises consistent with what can be found in the data implies that makeup strategies lose most of their traction. A systematic comparison of the implications of alternative expectations formation processes would be desirable.

We find that non-standard measures would not be sufficient in themselves to offset the destabilizing effects of the lower bound. We leave to future research to assess the robustness of this finding across alternative modelling choices of the channels of nonstandard measures.

References

- ADAM, K. AND R. M. BILLI (2006): "Optimal monetary policy under commitment with a zero bound on nominal interest rates," *Journal of Money, Credit and Banking*, 38, 1877–11905.
- ADJEMIAN, S. AND M. JUILLARD (2016): "Stochastic Extended Path," Mimeo.
- ADOLFSON, M., S. LASEEN, J. LINDE, AND L. E. SVENSSON (2011): "Optimal Monetary Policy in an operational medium-sized DSGE model," *Journal of Money, Credit and Banking*, 43, 1287–1331.
- ALTAVILLA, C., W. LEMKE, T. LINZERT, J. TAPKING, AND J. VON LANDESBERGER (2021): "Assessing the efficacy, efficiency and potential side effects of the ECB's monetary policy instruments since 2014," Occasional Paper Series 278, European Central Bank.
- AMANO, R. A., S. GNOCCHI, S. LEDUC, AND J. WAGNER (2020): "Average is good enough: Average-inflation targeting and the ELB," Tech. Rep. 2020-31, Bank of Canada Staff Working Paper.
- BERNANKE, B. S., M. T. KILEY, AND J. M. ROBERTS (2019): "Monetary policy strategies for a low-rate environment," in *AEA Papers and Proceedings*, vol. 109, 421–26.
- BODENSTEIN, M., J. HEBDEN, AND F. WINKLER (2019): "Learning and misperception: implications for Price-Level Targeting," Finance and Economics Discussion Series 2019-078, Board of Governors of the Federal Reserve System (U.S.).
- BRAND, C., M. BIELECKI, AND A. PENALVER (2018): "The natural rate of interest: estimates, drivers, and challenges to monetary policy," Occasional Paper Series 217, European Central Bank.
- BRAND, C. AND F. MAZELIS (2019): "Taylor-rule consistent estimates of the natural rate of interest," Working Paper Series 2257, European Central Bank.
- BUDIANTO, F., T. NAKATA, AND S. SCHMIDT (2020): "Average inflation targeting and the interest rate lower bound," Working Paper Series 2394, European Central Bank.
- BUSETTI, F., S. NERI, A. NOTARPIETRO, AND M. PISANI (2021): "Monetary policy strategies in the New Normal: a model-based analysis for the euro area," *Journal of Macroeconomics*, 70, 1–17.
- CHUNG, H. T., E. GAGNON, T. NAKATA, M. O. PAUSTIAN, B. SCHLUSCHE, J. TREVINO, D. VILÁN, AND W. ZHENG (2019): "Monetary policy options at the effective lower bound: assessing the Federal Reserve's current policy toolkit," Finance and Economics Discussion Series 3, Board of Governors of the Federal Reserve System.
- COENEN, G., C. MONTES-GALDÓN, AND S. SCHMIDT (2021): "Macroeconomic stabilisation and monetary policy effectiveness in a low-interest rate environment," *Journal* of Economic Dynamics and Control, 132, 1–14.

- COENEN, G., C. MONTES-GALDÓN, AND F. SMETS (2020): "Effects of state-dependent forward guidance, large-scale asset purchases and fiscal stimulus in a low-interest-rate environment," Working Paper Series 2352, European Central Bank.
- CORBO, V. AND I. STRID (2020): "MAJA: A two-region DSGE model for Sweden and its main trading partners," Working Paper Series 391, Sveriges Riksbank (Central Bank of Sweden).
- DE GROOT, O. AND F. MAZELIS (2020): "Mitigating the forward guidance puzzle: inattention, credibility, finite planning horizons and learning," Working Paper Series 2426, European Central Bank.
- DE GROOT, O., F. MAZELIS, R. MOTTO, AND A. RISTINIEMI (2021a): "A Toolkit for Computing Constrained Optimal Policy Projections (COPPs)," Working Paper Series 2555, European Central Bank.
- (2021b): "Estimating the degree of inattention to forward guidance announcements," Mimeo.
- DE REZENDE, R. AND A. RISTINIEMI (2023): "A shadow rate without a lower bound constraint," *Journal of Banking and Finance*, 146.
- DEL NEGRO, M., M. GIANNONI, AND C. PATTERSON (2015): "The forward guidance puzzle," Staff Reports 574, Federal Reserve Bank of New York.
- EGGERTSSON, G. B. AND M. GIANNONI (2020): "Medium-Term Money Neutrality and the Effective Lower Bound," Tech. rep., National Bureau of Economic Research.
- EGGERTSSON, G. B. AND M. WOODFORD (2003): "The zero lower bound on interest rates and optimal monetary policy," *Brookings Papers on Economic Activity*, 1, 139–211.
- ERCEG, C. J., Z. JAKAB, AND J. LINDÉ (2021): "Monetary policy strategies for the European Central Bank," *Journal of Economic Dynamics and Control*, 132, 104211.
- FAIR, R. AND J. TAYLOR (1983): "Solution and maximum likelihood estimation of dynamic nonlinear rational expectations models," *Econometrica*, 51, 1169–85.
- HARRISON, R. (2017): "Optimal quantitative easing," Bank of England working papers 678, Bank of England.
- HEBDEN, J. AND J. D. LÓPEZ-SALIDO (2018): "From Taylor's Rule to Bernanke's Temporary Price Level Targeting," Finance and Economics Discussion Series 2018-051, Board of Governors of the Federal Reserve System (U.S.).
- HOLDEN, T. D. (2016): Computation of solutions to dynamic models with occasionally binding constraints, Kiel und Hamburg: ZBW-Deutsche Zentralbibliothek für Wirtschaftswissenschaften.
- HOLSTON, K., T. LAUBACH, AND J. C. WILLIAMS (2017): "Measuring the natural rate of interest: International trends and determinants," *Journal of International Economics*, 108, 59–75.

- HONKAPOHJA, S. AND M. KAUSHIK (2020): "Price level targeting with evolving credibility," *Journal of Monetary Economics*, 116, 88–103.
- KILEY, M. T. (2013): "Output gaps," Journal of Macroeconomics, 37, 1-18.
- KILEY, M. T. AND J. M. ROBERTS (2017): "Monetary policy in a low interest rate world," *Brookings Papers on Economic Activity*, 48, 317–396.
- LEVIN, A., V. WIELAND, AND J. C. WILLIAMS (1999): Robustness of Simple Monetary Policy Rules under Model Uncertainty, University of Chicago Press.
- (2003): "The performance of forecast-based monetary policy rules under model uncertainty," *American Economic Review*, 93, 622–645.
- MAIH, J., F. MAZELIS, R. MOTTO, AND A. RISTINIEMI (2021): "Asymmetric monetary policy rules for the euro area and the US," *Journal of Macroeconomics*, 70.
- NAKATA, T., R. OGAKI, S. SCHMIDT, AND P. YOO (2019): "Attenuating the forward guidance puzzle: implications for optimal monetary policy," Working Paper Series 2220, European Central Bank.
- NAKOV, A. (2008): "Optimal and Simple Monetary Policy Rules with Zero Floor on the Nominal Interest Rate," *International Journal of Central Banking*, 4, 73–127.
- ORPHANIDES, A. AND V. WIELAND (1998): "Price stability and monetary policy effectiveness when nominal interest rates are bounded at zero," Finance and Economics Discussion Series 1998-35, Board of Governors of the Federal Reserve System.
- REIFSCHNEIDER, D. AND J. C. WILLIAMS (2000): "Three lessons for monetary policy in a low-inflation era," *Journal of Money, Credit and Banking*, 936–966.
- SCHMITT-GROHE, S. AND M. URIBE (2007): "Optimal simple and implementable monetary and fiscal rules," *Journal of Monetary Economics*, 54, 1702–1725.
- SMETS, F. AND R. WOUTERS (2007): "Shocks and Frictions in US Business Cycles: A Bayesian DSGE Approach," American Economic Review, 97, 586–606.
- SVENSSON, L. E. (2020): "Monetary policy strategies for the Federal Reserve," International Journal of Central Banking, 16, 133–193.
- VESTIN, D. (2006): "Price-level versus inflation targeting," Journal of Monetary Economics, 53, 1361–1376.
- WILLIAMS, J. C. (2003): "Simple rules for monetary policy," *Economic Review*, 1–12.
- ——— (2009): "Heeding Daedalus: optimal inflation and the zero lower bound," *Brookings* Papers on Economic Activity, 1–37.

Appendix

A The model

A.1 Model equations

With both a unit-root labour-augmenting technology shock, Z_t , and a stationary technology shock, ε_t^{α} , the production function of the intermediate goods producers is:

$$Y_t = \varepsilon_t^a (K_t^s(i))^\alpha (Z_t L_t(i))^{1-\alpha} - Z_t \Phi$$
(20)

with technological progress growing at the rate: $ln(Z_t/Z_{t-1}) = g_{z,t} = \rho^{g_z}g_{z,t-1} + \eta^{g_z}_t$.

All real variables in the model are detrended by the stochastic trend Z_t to ensure that the variables follow a balanced growth path:

$$y_t = Y_t/Z_t, \quad c_t = C_t/Z_t, \quad i_t = I_t/Z_t, \quad \bar{k}_t = \bar{K}_t/Z_t, \quad \tilde{p}_t = \tilde{P}_t/P_t, \quad \tilde{\omega}_t = \tilde{W}_t/W_t, \\ q_t^k = Q_t^k \quad k_t = K_t/Z_t, \quad \omega_t = W_t/(P_tZ_t), \quad r_t^k = R_t^k/P_t, \quad \xi_t = \Xi_t Z_t^{\sigma_c}$$

The production function of the intermediate goods producers becomes:

$$y_t Z_t = \varepsilon_t^a (k_t^s(i) Z_t)^\alpha (Z_t L_t(i))^{1-\alpha} - Z_t \Phi$$
(21)

Dividing throughout by Z_t :

$$y_t = \varepsilon_t^a (k_t^s(i))^\alpha (L_t(i))^{1-\alpha} - \Phi$$
(22)

The log-linearised model equations, where the real variables are detrended, are shown below.

Resource constraint:

$$\hat{y}_t = \hat{g}_t + c_y \hat{c}_t + i_y \hat{i}_t + z_y \hat{u}_t$$
(23)

Consumption Euler equation:

$$\hat{c}_{t} = \frac{1}{1 + he^{-g_{z}^{*}}} E_{t}[\hat{c}_{t+1}] + \frac{he^{-g_{z}^{*}}}{1 + he^{-g_{z}^{*}}} \hat{c}_{t-1} - \frac{(1 - he^{-g_{z}^{*}})}{\sigma_{c}(1 + he_{-g_{z}^{*}})} R_{t}^{gap} + \hat{b}$$
(24)
$$- \frac{1}{1 + he^{-g_{z}^{*}}} (he^{-\hat{g}_{z}^{*}}g_{z,t} - E_{t}\hat{g}_{z,t+1})$$
$$+ \frac{(\sigma_{c} - 1)}{\sigma_{c}(1 + he^{-g_{z}^{*}})} \frac{\omega^{*h}L^{*}}{c^{*}} (\hat{L}_{t} - E_{t}[\hat{L}_{t+1})]$$

Investment Euler equation:

$$\hat{i}_{t} = \frac{1}{S'' e^{2g_{z}^{*}} (1 + \beta e^{(1 - \sigma_{c})g_{z}^{*}})} \hat{q}_{t}^{k} + \frac{1}{1 + \beta e^{(1 - \sigma_{c})g_{z}^{*}}} (\hat{i}_{t-1} - \hat{g}_{z,t}) + \frac{\beta e^{(1 - \sigma_{c})g_{z}^{*}}}{1 + \beta e^{(1 - \sigma_{c})g_{z}^{*}}} E_{t} [\hat{i}_{t+1} + \hat{g}_{z,t+1}] + \mu_{t}$$

$$\tag{25}$$

Arbitrage equation value of capital:

$$\hat{q}_t^k = -R_t^{gap} + \frac{1}{\frac{1-he^{-g_z^*}}{\sigma_c(1+he^{-g_z^*})}}\hat{\hat{b}}_t + \frac{r^{*k}}{r^{*k} + (1-\delta)}E_t[r_{t+1}^k] + \frac{(1-\delta)}{r^{*k} + (1-\delta)}E_tq_{t+1}^k$$
(26)

Aggregate production function:

$$\hat{y}_t = \Phi(\varepsilon_t^a + \alpha \hat{k}_t + (1 - \alpha) \hat{L}_t)$$
(27)

Definition of capital services:

$$ln(k_t) - ln(k^*) = \hat{k}_t = \hat{u}_t - \hat{g}_{z,t} + \hat{\bar{k}}_{t-1}$$
(28)

First order condition, capacity utilisation:

$$u_t = \frac{1 - \psi}{1 - \psi} \hat{r}_t^k \tag{29}$$

Law of motion of capital:

$$\hat{\bar{k}}_t = (1 - \frac{i^*}{\bar{k}^*})(\hat{\bar{k}}_{t-1} - \hat{g}_{z,t}) + \frac{i^*}{\bar{k}^*}\hat{i}_t + \frac{i^*}{\bar{k}^*}S''e^{2g_z^*}(1 + \beta e^{(1 - \sigma_c)g_z^*})\hat{\mu}_t$$
(30)

First order condition, labour:

$$\hat{mc}_t = (1 - \alpha)\hat{\omega}_t + \alpha\hat{r}_t^k - \varepsilon_t^a \tag{31}$$

Price Phillips curve:

$$\hat{\pi}_{t} = \left(\frac{1}{1+\iota_{p}\bar{\beta}e^{g_{z}^{*}}}\right) \left(\bar{\beta}e^{g_{z}^{*}}E_{t}(\hat{\pi}_{t+1}) + \iota_{p}\hat{\pi}_{t-1} + (1-\iota_{p})\hat{\pi}_{t} - \bar{\beta}e^{g_{z}^{*}}(1-\iota_{p})E_{t}[\hat{\pi}_{t+1}] + \frac{1}{((\phi_{p}-1)\varepsilon_{p}+1)} \frac{(1-\zeta_{p})(1-\zeta_{p}\bar{\beta}e^{g_{z}^{*}})}{\zeta_{p}}\right)\hat{m}c_{t}) + \varepsilon_{p,t}$$
(32)

Firm first order condition, capital:

$$\hat{r}_t^k = \hat{\omega}_t + \hat{L}_t - \hat{k}_t \tag{33}$$

Wage Phillips curve:

$$\hat{\omega}_{t} = \frac{1}{1 + \bar{\beta}e^{g_{z}^{*}}} \left(\hat{\omega}_{t-1} - \hat{g}_{z,t} + \bar{\beta}e^{g_{z}^{*}} (E_{t}[\hat{\omega}_{t+1}] + \hat{g}_{z,t+1} + E_{t}[\hat{\pi}_{t+1}]) - (1 + \bar{\beta}e^{g_{z}^{*}}\iota_{\omega})\hat{\pi}_{t} + \iota_{\omega}\hat{\pi}_{t-1} + (1 - \iota_{\omega})\hat{\pi}_{t} - \bar{\beta}e^{g_{z}^{*}}(1 - \iota_{\omega})E_{t}[\hat{\pi}_{t+1}] + \frac{(1 - \zeta_{\omega}\bar{\beta}e^{g_{z}^{*}})(1 - \zeta_{\omega})}{\zeta_{\omega}((\phi_{\omega} - 1)\varepsilon_{\omega} + 1)} \left[-\hat{\omega}_{t} + \sigma_{l}\hat{L}_{t} + \frac{1}{1 - he^{-g_{z}^{*}}}\hat{c}_{t} - he^{-g_{z}^{*}}\hat{c}_{t-1} + he^{-g_{z}^{*}}\hat{g}_{z,t} \right] + \varepsilon_{\omega,t}^{\omega}$$

$$(34)$$

A.2 Shock processes:

$$\varepsilon_t^a = \rho^a \varepsilon_{t-1}^a + \eta_t^a \tag{35}$$

$$\varepsilon_t^{gz} = \rho^{gz} + \varepsilon_{t-1}^{gz} + \eta_t^{gz} \tag{36}$$

$$\varepsilon_t^b = \rho^b \varepsilon_{t-1}^b + \eta_t^b \tag{37}$$

$$\varepsilon_t^g = \rho^g \varepsilon_{t-1}^g + \eta_t^g \tag{38}$$

$$\varepsilon_t^i = \rho^i + \varepsilon_{t-1}^i + \eta_t^i \tag{39}$$

$$\varepsilon_t^m = \rho^m \varepsilon_{t-1}^m + \eta_t^m \tag{40}$$

$$\varepsilon_t^p = \rho \varepsilon_{t-1}^p + \eta_t^p + \mu_t^p \eta_{t-1}^p \tag{41}$$

$$\varepsilon_t^{\omega} = \rho \varepsilon_{t-1}^{\omega} + \eta_t^{\omega} + \mu_t^{\omega} \eta_{t-1}^{\omega}$$
(42)

$$\varepsilon_t^n = \rho^n \varepsilon_{t-1}^n + \nu_t^n \tag{43}$$

A.3 Data

We use eight quarterly euro area variables: the log difference of real GDP, real consumption, real investment, real wages, and the harmonised index of consumer prices; the log of hours worked; the short-term nominal interest rate; the output gap. We treat the output gap, y_t , as observable because we want it to be consistent with policymakers' view. We rely on the OECD's estimate of the output gap. For the interest rate we use the shadow rate constructed in De Rezende and Ristiniemi (2023). The measurement equations are shown below. Given uncertainty in measuring potential output, we have included a measurement error in the observation equation for the output gap.

$$dc_{t}^{obs} = c_{t} - c_{t-1} + g_{z,t} + g_{z}^{*}$$

$$di_{t}^{obs} = i_{t} - i_{t-1} + g_{z,t} + g_{z}^{*}$$

$$dy_{t}^{obs} = y_{t} - y_{t-1} + g_{z,t} + g_{z}^{*}$$

$$dw_{t}^{obs} = w_{t} - w_{t-1} + g_{z,t} + g_{z}^{*}$$

$$y_{t}^{obs} = y_{t} + \sigma^{y} \varepsilon_{t}^{y}$$

$$pi_{t}^{obs} = \pi_{t} + \bar{\pi}$$

$$l_{t}^{obs} = l_{t} + \bar{l}$$

$$R_{t}^{obs} = R_{t} + R^{*}$$

B Calibration

Parameter		Calibration
Steady state inflation rate (quarterly)	$\bar{\pi}$	0.475
Capital depreciation rate	δ	0.025
Gross mark-up on wages	ϕ_w	1.3
Share of government spending in output	$rac{ar{g}}{ar{y}}$	0.18
Curvature of Kimball aggregator for wages	ε_w	10
Curvature of Kimball aggregator for prices	ε_p	10
Discount rate	$100(\beta^{-1}-1)$	0.16
Measurement error of standard deviation of the output gap	σ^y	0.285
QE on shadow rate	ξ^{qe}	0.02
QE autoregressive parameter, t-1	$ ho^{qe1}$	1.5
QE autoregressive parameter, t-2	$ ho^{qe2}$	-0.54
QE parameter on inflation	$ heta^{qe}$	-2.5
QE parameter on quarterly Taylor interest rate gap	$ au^{qe}$	1.25
		1

 Table 2: Calibrated parameters

C Estimation

			Pr	ior	Posterior
	Variable	Distribution	Mean	Std	Mode
Standard errors:					
Technology shock	σ_a	Inv. Gamma	0.1	2	0.25
Risk premium shock	σ_b	Inv. Gamma	0.1	2	0.05
Government spending shock	σ_q	Inv. Gamma	0.1	2	0.32
Investment shock	σ_i	Inv. Gamma	0.1	2	0.95
Monetary policy shock	σ_m	Inv. Gamma	0.1	2	0.13
Price mark-up shock	σ_p	Inv. Gamma	0.1	2	0.03
Wage mark-up shock	σ_w	Inv. Gamma	0.1	2	0.13
Permanent technology shock	σ_{gz}	Inv. Gamma	0.1	2	0.16
Persistence:	5				
Technology shock	ρ^a	Beta	0.5	0.2	0.91
Risk premium shock	ρ^b	Beta	0.8	0.1	0.90
Government spending shock	ρ^g	Beta	0.5	0.2	0.97
Investment shock	ρ^i	Beta	0.5	0.2	0.08
Monetary policy shock	ρ^m	Beta	0.5	0.2	0.65
Price mark-up shock	ρ^p	Beta	0.5	0.2	0.85
Wage mark-up shock	$ ho^{\omega}$	Beta	0.5	0.2	0.72
Permanent technology shock	ρ^{gz}	Beta	0.5	0.05	0.52
Other parameters:		I			
Calvo wages	ξ_w	Beta	0.75	0.1	0.84
Calvo prices	ξ_p	Beta	0.75	0.02	0.77
Indexation wages	ι_w	Beta	0.5	0.2	0.33
Indexation prices	ι_p	Beta	0.5	0.2	0.75
Coeff. MA term price mark-up	μ_p	Beta	0.5	0.2	0.25
Coeff. MA term wage mark-up	μ_w	Beta	0.5	0.2	0.91
Risk aversion	σ_c	Gaussian	1.5	0.1	1.21
Frisch elasticity	σ_l	Gaussian	2.5	0.75	2.88
Degree of habit formation	λ	Beta	0.7	0.1	0.79
Investment adjustment cost	φ	Gaussian	6	1.5	7.00
Fixed cost share	ϕ_p	Gaussian	1.25	0.125	1.36
Capacity utilisation cost	-	Beta	0.5	0.1	0.55
Capital share	α	Gaussian	0.3	0.05	0.11
Steady state net growth	$100(g_z^* - 1)$	Gaussian	0.3	0.02	0.28
Steady state hours	ī	Gaussian	0	2	0.83
Coeff. Potential growth on R^*	θ^{rn}	Beta	0.15	0.1	0.08
Feedback tech. on ex. spending	$ ho_{ga}$	Gaussian	0.5	0.2	0.69
Policy rule inflation	r_{π}	Gaussian	1.85	0.25	2.12
Policy rule output gap	r_y	Gaussian	0.1	0.2	0.30
Policy rule persistence	ρ^m	Beta	0.75	0.1	0.79

 Table 4: Priors and posteriors

D Simulations



Figure 9: Comparison of methods to adjust the natural rate of interest

Note: The simulations are conducted with DynareOBC (Holden, 2016). The same random shock series is applied to all different specifications.

			Lower-bound dence	nd inci-	Inflation	Inflation (P.P., annual)	Outpu	Output gap (%)	Growth (%, quarterly)	Interest ra	Growth (%, quarterly) Interest rate (P.P., annual)
Inflation aim pi*	Equilibrium real in- Effective terest rate r^* (p.a.) bound (p.a.)	a.)	lower Frequency (%)	Avg. du- ration (quar- ters)	. Mean	Std	Mean	Std	Mean	Std	M ean	Std
		0.00%	10.63	16.5	2.9	2.16	-0.28	3.69	-0.01	2.56	4.17	2.81
3%	1%	-0.50%	16.7	14.2	2.94	2.06	-0.18	3.39	0	2.4	4.11	2.87
	2	0.00%	11.09	16.95	1.92	2.22	-0.29	3.8	0	2.65	4.15	2.83
	2%	-0.50%	8.38	14.91	1.95	2.07	-0.2	3.4	0	2.47	4.09	2.89
		0.00%	18.04	22.52	1.74	3.05	-0.75	5.81	0	3.7	3.37	2.7
2%	. 1%	-0.50%	14.58	19.93	1.84	2.57	-0.49	4.7	-0.01	3.08	3.23	2.77
		0.00%	23.84	26.54	1.55	4.15	-1.25	8.13	0.01	4.92	2.87	2.56
	%0	-0.50%	21.11	24.52	1.64	3.44	-0.98	6.68	0	4.21	2.58	2.62
2	2	0.00%	28.1	30.11	0.45	4.41	-1.45	8.48	0.05	5.33	2.58	2.46
1%	1%	-0.50%	26.13	28.23	0.52	3.85	-1.29	7.46	0	4.82	2.17	2.48

 Table 5: ELB simulation results

Note: The statistics are based on 2000 simulations of 150 periods, excluding the first 50 periods as burn-in. We exclude runs where inflation goes below – 10%, as this substantially biases the results.

	Lower-boun dence	Lower-bound inci- dence	Inflation	Inflation (annual)	Output gap (%)	gap (%)
Policy rules	Frequency (%)	Average du- ration (quar- ters)	Mean	Std	Mean	Std
Inflation targeting rule	18.02	22.53	1.74		-0.74	5.79
Inflation targeting rule + QE-rule only at the ELB	19.01	23.41	1.88		-0.47	3.81
Inflation targeting rule $+$ permanent QE-rule	10.91	16.88	1.99	1.08	-0.08	3.23
Inflation targeting rule + QE-rule only at the ELB, QE max 50% of long-term bonds		20.96	1.91		-0.36	4.00

 Table 6: Asset purchase simulation results

Note: The statistics are based on 2000 simulations of 150 periods, excluding the first 50 periods as burn-in. We exclude runs where inflation goes below -10%, as this substantially biases the results. The effective lower bound of the policy rate is set to -0.5%. The inflation target π^* is set to 2% and $r^* = 0.5\%$.

		Lower-bound in- cidence	und in-	Inflatio	Inflation (annual)	Output	Output gap (%)	Welfare
Policy rules		Frequency (%)	Average duration (quar- ters)	Mean	Std	Mean 	Std	Loss
	with ELB of -0.5%	18.02	22.53	1.74	3.04	-0.74	5.79	18.02
Benchmark inflation targeting rule	without ELB			2.02	1.77	0.00	2.50	4.89
4-year Average Inflation Targeting rule		16.28	23.01	1.80	2.85	-0.57	6.22	18.04
8-year Average Inflation Targeting rule		14.27	20.21	1.97	1.39	-0.21	4.60	7.35
Price Level Targeting		13.34	19.30	1.99	3.48	-0.17	7.49	26.29
Nominal GDP Targeting		17.20	22.01	2.00	1.44	-0.21	4.05	6.35
Asymmetric inflation rule, $\phi = 1$		19.65	24.03	1.58	4.81	-1.12	9.53	23.83

Table 7: Make-up strategy simulation results

Note: The statistics are based on 2000 simulations of 150 periods, excluding the first 50 periods as burn-in. We also exclude runs where inflation goes below -10 as this substantially biases the results. The effective lower bound of the policy rate is set to -0.5%. The inflation target π^* is set to 2% and $r^* = 0.5\%$. The Loss is computed using Equation (16).

Acknowledgements

We are grateful for comments by Eurosystem colleagues in the context of the 2021 ECB strategy review's work stream on the price stability objective and thank Stephane Adjemian for discussions on simulation methods, as well as Camilo Marchesini and Maria Giulia Cassinis for excellent research assistance.

The views expressed herein do not necessarily reflect those of the European Central Bank or the Eurosystem.

Falk Mazelis

European Central Bank, Frankfurt am Main, Germany; email: Falk.Mazelis@ecb.europa.eu

Roberto Motto

European Central Bank, Frankfurt am Main, Germany; email: roberto.motto@ecb.europa.eu

Annukka Ristiniemi

European Central Bank, Frankfurt am Main, Germany; email: Annukka.Ristiniemi@ecb.europa.eu

© European Central Bank, 2023

Postal address 60640 Frankfurt am Main, Germany Telephone +49 69 1344 0 Website www.ecb.europa.eu

All rights reserved. Any reproduction, publication and reprint in the form of a different publication, whether printed or produced electronically, in whole or in part, is permitted only with the explicit written authorisation of the ECB or the authors.

This paper can be downloaded without charge from www.ecb.europa.eu, from the Social Science Research Network electronic library or from RePEc: Research Papers in Economics. Information on all of the papers published in the ECB Working Paper Series can be found on the ECB's website.

PDF	ISBN 978-92-899-5993-3	ISSN 1725-2806	doi:10.2866/187903	QB-AR-23-034-EN-N